Author’s response to reviews

Title: EPIDEMIOLOGICAL AND HISTOPATHOLOGICAL PROFILE OF MALIGNANT MELANOMA IN MALAWI

Authors:

Maurice Mulenga (Maurice.Mulenga1@gmail.com)
Nathan Montgomery (Nathan.Montgomery@unchealth.unc.edu)
Maganizo Chagomerana (mchagomerana@unclilongwe.org)
Tenganawo Mzumala (estherkafwafwa@gmail.com)
Tamiwe Tomoka (ttomoka@unclilongwe.org)
Coxcilly Kampani (ckampani@unclilongwe.org)
Yuri Fedoriw (Yuri.Fedoriw@unchealth.unc.edu)
Satish Gopal (Satish_gopal@med.unc.edu)
Lisbet Sviland (lisbet.sviland@helse-bergen.no)

Version: 4 Date: 21 Feb 2019

Author’s response to reviews:

The specific individual contributions of ALL authors to the manuscript should be specified in the Author’s Contributions section. Currently authors Nathan Montgomery (NM), Tenganawo Mzumala (TM) and Yuri Fedoriw (YF) are missing. Please amend this.

*I included the above mentioned authors in the last statement under the Author’s Contributions section that read “All authors contributed to the interpretation of the results, edited and approved the final manuscript”.

*However for clarity, I have deleted the sentence and replaced it with this one “MM, NC, MC, TM, TT, CK, YF, SG and LS interpreted the results, edited, substantively revised and approved the final manuscript”.